Author's response to reviews

Title: Development and implementation of a mini-Clinical Evaluation Exercise (mini-CEX) program to assess the clinical competencies of internal medicine residents: from faculty development to curriculum evaluation

Authors:

Kuo-Chen Liao (kuochenliao@yahoo.com.tw)
Shou-Jin Pu (shoujinpu@adm.cgmh.org.tw)
Maw-Sen Liu (sam0033@adm.cgmh.org.tw)
Chih-Wei Yang (cwyang@adm.cgmh.org.tw)
Han-Pin Kuo (hpkuo100@adm.cgmh.org.tw)

Version: 3 Date: 6 February 2013

Author's response to reviews: see over
February 5, 2012

BMC Medical Education

RE: Manuscript ID 1119873634738078
Title: "The development and implementation of a mini-Clinical Evaluation Exercise (mini-CEX) program to assess the clinical competencies of internal medicine residents: From faculty development to curriculum evaluation"

Dear Editor,

We respectfully submit our latest revised manuscript entitled “Development and implementation of a mini-Clinical Evaluation Exercise (mini-CEX) program to assess the clinical competencies of internal medicine residents: from faculty development to curriculum evaluation”.

We thank the editor for the comment. The typographical and grammatical errors were carefully corrected. We hope the revised manuscript will finally meet the standards of the “BMC Medical Education”.

Again, we would like to express our gratitude for the critical review provided by the reviewers and editor and truly thank for their time and consideration.

Sincerely yours,

Kuo-Chen Liao, MD
Division of General Medicine and Geriatrics,
Department of Internal Medicine, Chang Gung Memorial Hospital
Chang Gung University, College of Medicine
5 Fusing street, Gueishan, Taoyuan 333, Taiwan
Tel: 886-3-3281200 ext 8430
Fax: 886-3-3281200 ext 8444
E-mail: kuochenliao@yahoo.com.tw
Response to Editor’s comment

"Minor amendments - copy editing needed. Still some typographical and grammatical errors. The authors should be able to do this. Once the corrections are made, the manuscript should be published."

**Response:** The typographical and grammatical errors were corrected carefully in the revised manuscript.